Author's response to reviews

Title: Atypical clinical presentation of Mucopolysaccharidosis Type II, Hunter syndrome.

Authors:

Gauri S Shah (gaurishankarshah@hotmail.com)
Tania Mahal (tania@hotmail.com)
Subodh Sharma (seesubodh@hotmail.com)

Version: 3 Date: 26 January 2010

Author's response to reviews: see over
Title: Atypical clinical presentation of Mucopolysaccharidosis Type II, Hunter Syndrome.

Shah GS 1, Tania Mahal 2, Subodh Sharma 3,

Gauri Shankar Shah 1

Tania Mahal 2

Subodh Sharma 3

1 Associate Professor
Department of Pediatrics and Adolescent Medicine
B P Koirala Institute of Health Sciences, Dharan, Nepal
gaurishankarshah@hotmail.com

2 Junior Resident
Department of Pediatrics and Adolescent Medicine
B P Koirala Institute of Health Sciences, Dharan, Nepal
tania@hotmail.com

3 Junior Resident
Department of Pediatrics and Adolescent Medicine
B P Koirala Institute of Health Sciences, Dharan, Nepal
seessubodh@hotmail.com

Affiliation:

From the Department of Pediatrics and Adolescent Medicine, B.P. Koirala Institute of Health Sciences, Dharan, Nepal.

Address for correspondence:
Dr. Gauri Shankar Shah
Associate professor
Department of Paediatrics and Adolescent Medicine
B P Koirala Institute of Health sciences, Dharan, Nepal
Email: gaurishankarshah@hotmail.com
Funding: None

Competing Interest: None

Contributors: Gauri Shankar Shah conceived and designed the study and wrote the initial draft of the manuscript. He shall act as the guarantor. Tania Mahal and Subodh Sharma collected the data and drafted the paper. Gauri Shankar Shah helped in acquisition and revised the manuscript for important intellectual content. The final manuscript was approved by all authors.

Written informed consent was obtained from the patient for publication of this case report and accompanying images.